



Expert Review of Clinical Pharmacology

ISSN: 1751-2433 (Print) 1751-2441 (Online) Journal homepage: http://www.tandfonline.com/loi/ierj20

Disclosing negative trial results – procedure

Cristina Scavone, Concetta Rafaniello, Liberata Sportiello, Liberato Berrino & Annalisa Capuano

To cite this article: Cristina Scavone, Concetta Rafaniello, Liberata Sportiello, Liberato Berrino & Annalisa Capuano (2016): Disclosing negative trial results – procedure, Expert Review of Clinical Pharmacology, DOI: 10.1080/17512433.2016.1251838

To link to this article: http://dx.doi.org/10.1080/17512433.2016.1251838

Accepted author version posted online: 25 Oct 2016. Published online: 02 Nov 2016.



🕼 Submit your article to this journal 🗗

Article views: 58



View related articles 🗹



View Crossmark data 🗹

Full Terms & Conditions of access and use can be found at http://www.tandfonline.com/action/journalInformation?journalCode=ierj20

EDITORIAL



Disclosing negative trial results - procedure

Cristina Scavone, Concetta Rafaniello, Liberata Sportiello, Liberato Berrino and Annalisa Capuano

Regional Center of Pharmacovigilance, Department of Experimental Medicine – Section of Pharmacology 'L. Donatelli', Second University of Naples, Naples (Italy)

ARTICLE HISTORY Received 29 August 2016; Accepted 19 October 2016

KEYWORDS Research; trial; negative; results; disclosure

1. Introduction

In order to identify and reduce biases in clinical research, during the last years, several rigorous methodological methods were applied to clinical trial management. Nevertheless, to date, too many biases still affect clinical studies. Since 1986, when the first data demonstrating publication bias was published, several biases related to clinical trial management were identified [1]. One of misconduct clinical research is also identified in reporting results. The so-called 'file-drawer effect' refers to the non-reporting of clinical trial results, which are often negative or neutral [2]. During the last 15 years, several Big Pharmas withheld negative trial results, such as the cases of the cardiovascular toxicity associated with rofecoxib and suicide-related behavior associated with paroxetine [3]. Unpublished negative trial results are not the only problem. Sometimes, also positive trial results are fake or exaggerated [4]. An example of withholding negative information is represented by the case of a celecoxib CLASS trial. Author's trial reported a better safety for celecoxib compared to ibuprofen and diclofenac at 6 months of follow-up, but they did not show results at week 65, which demonstrated that the drugs were associated to a similar number of ulcer complications [5,6].

In the light of these considerations, the main risk factors for the omission on negative trial results and/or the publication of the false ones, such as the clinical trial characteristics, conflicts of interest, blinded-treatment, and definition of high quality endpoints, have to be changed. Among these factors, the conflict of interest, a set of conditions in which the judgment on a professional interest tends to be disproportionately influenced by secondary interest, has to be strictly controlled. Since conflict of interest could influence the management of a clinical trial and lead to several publication biases, especially those related to the reporting of efficacy and safety results, it would be desirable that the investigators had not any conflicts of interest or at least they had to declare it [7].

In order to minimize the risk associated to clinical trial unpublished negative results and/or exaggerated positive ones, avoiding the distortion of clinical evidence deriving from such activities, different approaches can be adopted. Many projects have been recently started (i.e. the European Medicine Agency proactive publication of clinical trial data [8] and the Ben Goldacre AllTrial initiative [9]), but a great help for the management and the publications of clinical studies performed in European Union (EU) countries was given by the adoption of the new Clinical Trials Regulation (CTR) No. 536/ 2014, which will be effective from 2018. One of the main aims of the new Regulation is to guarantee that EU Member States authorize and supervise the management of clinical trials on identical rules. Since transparency is undoubtedly the key value that promises the better access to all clinical trial data, a great importance is given to this aspect. According to CTR No. 536/2014, all the data regarding clinical trials application will be submitted in the EU database and will be publically accessible (except for commercially, personal, and communication confidential data). According to the new Regulation, the sponsor has to submit a summary of trial results within 1 year by the end of the trial. Moreover, 30 days after the marketing authorization, a Clinical Summary Report shall be submitted [10]. Although the undeniable usefulness of changes introduced by the new Regulation, we will have to wait years before all these will be real. Thus, other approaches, such as the potentiation of registration sites of clinical studies, the publication of Development Safety Update Report (DSUR) and Investigator's Brochure (IB), the promotion of independent clinical research, and the establishment for each negative trial of a section dedicated to an external expert, can be adopted.

2. Registration of clinical studies

In July 2005 the International Committee of Medical Journal Editors required the registration of all clinical trial before the publication of their results. However, a recent analysis of Harriman et al. revealed that out of 108 clinical studies published in 2013 on BMC journal, 33 and 72 were registered prospectively and retrospectively before the publication of their results, respectively, while 6 were registered after the publication on BMC [11]. Even though two public big databases are today available for the registration of clinical trial

CONTACT Annalisa Capuano annalisa.capuano@unina2.it Department of Experimental Medicine – Section of Pharmacology 'L. Donatelli', Second University of Naples, Naples, Italy

(ClinicalTrial.gov, instituted by Food and Drug Administration – FDA, and European union Drug Regulating Authorities Clinical Trials – EudraCT by EMA), researchers are still not prone to register prospectively their trial. The prospective registration of clinical trial data could potentially represent the way to avoid the omission of negative trial results and to improve clinical trial transparency and reduce publication biases.

3. Promoting the publication of DSUR

Given the limitation of premarketing studies, such as the shortterm drug exposure, the selection of patient populations, and the clinical trial duration, missing efficacy and safety data on medicines are frequently detected during the postmarketing phase. The main aim of the DSUR is to share with regulatory agencies, ethic committees, clinicians and, patients new safety data and findings from both premarketing and postmarketing studies of drugs under investigation. According to EMA ICH guideline E2F, DSUR is an annual report that should be submitted to regulatory authorities for as long as the sponsor manages clinical trials with the investigational drug. Considering that DSUR contains both cumulative and periodic safety information relating to the investigational drug, it will be more useful if this document could be accessible to anyone interested. In that way, new drug informations will be available for clinicians and patients and safer use of drugs will be allowed. Given the importance of data listed in the IB, this document should be available for everyone. However, since IB very frequently includes confidential information, according to Good Clinical Practice, the protection of rights, integrity, and confidentiality of trial subjects must be guaranteed [12].

4. Promoting the potentiation of independent clinical research

The results of a survey on 324 cardiovascular trials published between 2000 and 2005 revealed that clinical studies funded by profit organizations were more likely to report positive findings than trials funded by non-profit organizations. Specifically, the authors found that 67% of profit studies favored the newer treatment, while only 49% of non-profit studies did [13]. Since no-profit clinical studies are not aimed at the industrial development of the drug but at the improvement of proper use of medicine in 'real-world conditions', particularly in populations usually not involved in clinical trials (such as pediatrics, elderlies, and/or pregnant women, patients with comorbid conditions or patients under multiple drug therapies), there is an urgent need to improve this sector.

5. Comment of external expert on each negative trial

Data Monitoring Committees (DMC), introduced as components of clinical trials, were aimed to review the safety, effectiveness, and trial issues data. Similarly to DMC, a further approach, which can implement the right interpretation of all clinical trial data, is the establishment of a section dedicated to a external advisory expert board opinion. The involvement of external experts who are allowed to analyze and discuss each failed trial could potentially help the readers to understand how much confidence can be placed in the clinical trial results. Therefore, an external opinion for each clinical trial, especially for the negative ones, will help to understand the statistical power of a trial, monitor efficacy, for example, in terms of achievement of primary end point, and also safety, through the evaluation of adverse event rates. In this context, the independence of an external expert will more deeply define if the study results appropriately meet research standards.In conclusion, sharing clinical trial results, especially the negative ones, as well as statistical analysis and methodological data, is one of the main aims of the current clinical research. Sharing these data will generate several benefits, leading to a better understanding of benefit/risk profile of a new treatment [14]. On the contrary, a misconducting clinical trial can lead to inefficiency in research resource allocation as well as significant changes of the regulatory decision-making processes [15].

Benefits deriving from clinical trial data sharing are clear. The instruments in order to guarantee transparency of clinical study's characteristics as well as its efficacy and safety results are now available. Before the new CTR No. 536/2014 will be real, definitively effective, clinical trials registration, DSUR and IB publication, promotion of independent clinical research, and the possibility for an external expert to comment on each negative trial may represents important instruments which can guarantee clinical trial data transparency.

Funding

This paper was not funded.

Declaration of interest

The authors have no relevant affiliations or financial involvement with any organization or entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript. This includes employment, consultancies, honoraria, stock ownership or options, expert testimony, grants or patents received or pending, or royalties.

References

Papers of special note have been highlighted as either of interest (•) or of considerable interest (••) to readers.

- 1. Goldacre B. How to get all trial reported: audit, better data, and individual accountability. Plos Med. 2015;12:e1001821.
- •• Recent and important article of Goldacre B. on reporting clinical trial results.
- 2. Capuano A, Coats AJS, Scavone C, et al. Disclosure of negative trial results. A call for action. Int J Cardiol. 2015;198:47–48.
- Recent editorial on disclosure of negative trial results.
- 3. Kesselheim AS, Avorn J. The role of litigation in defining drug risks. Jama. 2007 Jan 17;297(3):308–311.

- 4. Ioannidis JP. How to make more published research true. PLoS Med. 2014 Oct 21;11(10):e1001747.
- Silverstein FE, Faich G, Goldstein JL, et al. Gastrointestinal toxicity with celecoxib vs nonsteroidal antinflammatory drugs for osteoarthritis and rheumatoid arthritis: the CLASS study: a randomized controlled trial. Jama. 2000;284:1247–1255.
- Hrachovec JB, Mora M. Reporting of 6-month vs 12-month data in a clinical trial of celecoxib. Jama. 2001 Nov 21;286(19):2398. author reply 2399-400.
- Lo B, Field MJ. Field. Institute of Medicine (US) Committee on Conflict of Interest in Medical Research, Education, and Practice. Available from: http://www.ncbi.nlm.nih.gov/books/ NBK22940/
- [cited 2016 Aug 2]. Available from: http://www.ema.europa.eu/ docs/en_GB/document_library/Report/2014/10/WC500174378.pdf.
- •• Questions and answers of European Medicines Agency on publication of clinical data for medicines.
- 9. [cited 2016 Aug 2]. Available from: http://www.alltrials.net/find-out -more/why-this-matters/the-alltrials-campaign/.
- 10. European Commission DG Health and Food Safety Public health Medicinal products for human use Clinical trials Regulation. [cited

2016 Aug 2]. Available from: http://ec.europa.eu/health/human-use /clinical-trials/regulation/index_en.htm

- 11. Harriman S, Patel J. When are clinical trials registered? an analysis of prospective versus retrospective registration. Trial. 2016;17:187.
- 12. ICH guideline E2F on development safety update report Step 5. [cited 2016 Aug 2]. Available from: http://www.ema.europa.eu/ docs/en_GB/document_library/Scientific_guideline/2010/09/ WC500097061.pdf.
- Ridker PM, Torres J. Reported outcomes in major cardiovascular clinical trials funded by for-profit and not-for-profit organizations: 2000-2005. Jama. 2006 May 17;295(19):2270–2274. Erratum in: JAMA. 2006 Jun 21;295(23):2726.
- Unger JM, Barlow WE, Ramsey SD, et al. The scientific impact of positive and negative phase 3 cancer clinical trials. JAMA Oncol. 2016;2:875–881.
- 15. Condivisione dei dati provenienti dalle sperimentazioni cliniche: l'importanza di un approccio globale. [cited 2016 Aug 2]. Available from: http://www.agenziafarmaco.gov.it/it/content/condi visione-dei-dati-provenienti-dalle-sperimentazioni-cliniche-l%E2% 80%99importanza-di-un-approcci).